Author’s response to reviews

Title: The isothiocyanate class of bioactive nutrients covalently inhibit the MEKK1 protein kinase

Authors:

Janet V Cross (cross@virginia.edu)
Frank W Foss Jr (ff2185@columbia.edu)
Joshua M Rady (jmr7k@virginia.edu)
Timothy L Macdonald (tlm@virginia.edu)
Dennis J Templeton (templeton@virginia.edu)

Version: 3  Date: 19 September 2007

Author’s response to reviews:

This revised manuscript contains all of the formatting revisions requested by the editorial staff. Revised figures have also been provided, to remove the numbers that were included on the earlier version. One additional change was made in the manuscript with the insertion of an acknowledgements section listing our funding source, that was inadvertently omitted from the earlier manuscript.